Author's response to reviews

Title: Biomonitoring of bisphenol A concentrations in maternal and umbilical cord blood in regard to birth outcomes and adipokine expression: a birth cohort study in Taiwan

Authors:

Wei-Chun Chou (d9712871@oz.nthu.edu.tw)
Jyh-Larng Chen (jlichen@mail.ypu.edu.tw)
Chung-Fen Lin (chungfen1119@yahoo.com.tw)
Yi-Chun Chen (cyc@mail.cjcu.edu.tw)
Feng-Cheng Shih (fengcheng@mail.ypu.edu.tw)
Chun-Yu Chuang (cychuang@mx.nthu.edu.tw)

Version: 4 Date: 7 October 2011

Author's response to reviews:

Thanks for these valuable comments greatly improving the quality of this manuscript. The manuscript has been revised with the comments carefully considered. Furthermore, the manuscript have been checked from native English language speaker.